Author's response to reviews

Title: Varicella Zoster Virus infection presenting as isolated diplopia: a case report

Authors:

Raffaella Pisapia (raffaella.pisapia@inmi.it)
Alessia Rianda (alessia.rianda@inmi.it)
Andrea Mariano (andrea.mariano@inmi.it)
Angela Testa (angela.testa@inmi.it)
Simonetta Galgani (sgalgani@scamilloforlanini.rm.it)
Laura Vincenzi (laura.vincenti@inmi.it)
Amina Abdeddaim (amina.abeddaim@inmi.it)
Alessandra Oliva (alessandra.oliva@inmi.it)
Pasquale Narciso (pasquale.narciso@inmi.it)

Version: 2 Date: 24 July 2012

Author's response to reviews:

Dear Editor, we modified the editorial form as requested. Thank you for the suggestions. Please find enclosed the revised version.

RP